Author's response to reviews

Title: Poor ability to resist tempting calorie rich food is linked to altered balance between neural systems involved in urge and self-control

Authors:

Qinghua He (heqinghua@gmail.com)
Lin Xiao (linxiao@usc.edu)
Gui Xue (guixue@gmail.com)
Savio Wong (savio.wong@gmail.com)
Susan L Ames (susan.ames@cgu.edu)
Susan M Schembre (SSchembre@mdanderson.org)
Antoine Bechara (bechara@usc.edu)

Version: 2
Date: 7 May 2014

Author's response to reviews:

Please find changes in red in the main text.
The email addresses for each author had been added to the title page.
Also, the IRB reference number has been added to the Methods section.